Author's response to reviews

Title: Hypertension Improvement Project (HIP): study protocol and implementation challenges

Authors:

Rowena J. Dolor (rowena.dolor@duke.edu)
William S. Yancy Jr. (yancy006@mc.duke.edu)
William F. Owen (wfowenmd@umdnj.edu)
David B. Matchar (david.matchar@duke.edu)
Gregory P. Samsa (samsa001@mc.duke.edu)
Kathryn I. Pollak (kathryn.pollak@duke.edu)
Pao-Hwa Lin (lin00004@mc.duke.edu)
Jamy D. Ard (jamy.ard@duke.edu)
Maxwell Prempeh (Maxwell.Prempeh@uphs.upenn.edu)
Heather L. McGuire (hmguire@billingsclinic.org)
Bryan C. Batch (bryan.batch@duke.edu)
William Fan (fanw73@yahoo.com)
Laura P. Svetkey (svetk001@mc.duke.edu)

Version: 3 Date: 19 December 2008

Author's response to reviews:

Per the request of Abigail Jones, we are re-uploading this manuscript, which has been reformatted from its original version in order to conform with the submission guidelines for study protocols.